Author's response to reviews

Title: Early life adversity and C-reactive protein in diverse populations of older adults: a cross-sectional analysis from the International Mobility in Aging Study (IMIAS)

Authors:

Annie Li (annie.li@mail.mcgill.ca)
Mai Thanh Tu (mai.thanh.tu@umontreal.ca)
Ana Carolina Sousa (acapas@gmail.com)
Beatriz Alvarado (alvaradbody@queensu.ca)
Georges Karna Kone (karnageorges@gmail.com)
Jack Guralnik (jguralnik@epi.umaryland.edu)
Maria Victoria Zunzunegui (maria.victoria.zunzunegui@umontreal.ca)

Version: 4 Date: 27 May 2015

Author's response to reviews: see over
Dear Dr. Volpato,

I am pleased to resubmit our revised manuscript, “Early life adversity and C-reactive protein in diverse populations of older adults: a cross-sectional analysis from the International Mobility in Aging Study (IMIAS)”, to BMC Geriatrics.

We have made the following improvements to the article:

-Comment:

“In both the abstract and in the results section the authors claim that in fully adjusted regression analysis, childhood social adversity was significantly associated with hsCRP. However, results presented in table 3 do not support this statement (p=.053). The authors should clarify this point.”

-Reply:

You are right. We have made corresponding changes to the abstract, results, discussion and conclusion.

Best regards,

Annie Li
MDCM
Department of Family Medicine, McGill University
5858, chemin de la Côte-des-Neiges, 3rd floor
Montreal, Quebec, H3S 1Z1, Canada
annie.li@mail.mcgill.ca